Author's response to reviews

Title: What ICU nurses in different Austrian hospitals know and think about the Austrian organ donation law

Authors:

Gabriele Zettel (gabriele.zettel@medunigraz.at)
Angela Horvath (angela.horvath@medunigraz.at)
Ekaterina Vorobyeva (ekatarina.vorobyeva@stud.medunigraz.at)
Christian Auburger (c.auburger@hotmail.com)
Michael Zink (michael.zink@bbstveit.at)
Philipp Stiegler (philipp.stiegler@medunigraz.at)
Vanessa Stadlbauer (vanessa.stadlbauer@medunigraz.at)

Version: 2 Date: 8 May 2014

Author's response to reviews:

Dear Dr. Jose

I have changed the wording from "authors roles" to "authors´ contributions" and added the sentence "All authors approved the final manuscript.

Kind regards

V. Stadlbauer (on behalf of the coauthors)